Author's response to reviews

Title: The Optimal Starting Time Of Postoperative Intraperitoneal Mitomycin-C Therapy With Preserved Intestinal Wound Healing

Authors:

Ali Uzunkoy (aliuzunkoy@yahoo.com)
Cengiz Bolukbas (drcengizbolukbas@hotmail.com)
Mehmet Horoz (mehmethoroz@yahoo.com)
Fusun F Bolukbas (fusunbol@yahoo.ca)
Abdurrahim Kocyigit (kocyigita@harran.edu.tr)

Version: 3 Date: 27 January 2005

Dear Editor,

Thanks for attention that you have paid to our manuscript entitled as "The Optimal Starting Time Of Postoperative Intraperitoneal Mitomycin-C Therapy With Preserved Intestinal Wound Healing (MS: 4076810250841512)". Please find attached revised manuscript. We revised the manuscript according to reviewer comments. The revised manuscript also conforms to all of the points provided at the manuscript formatting checklist. The revisions that made are listed below by providing with a point-by-point response.

Revisions:
1. All P values used at the results section were adjusted according to reviewer advice as p < or > 0.05.
2. We explained the mean of (both, p<0.001) that used at page 6 line 8, 10, 16, 17 of old version of the manuscript at results section. P values were expressed as p< 0.05 and cited at page 6 line 11, 14, 20 and 23 of revised manuscript.
3. The reason of exclusion of dead rats was explained at page 5 line 22, 23, 24, 25 and page 6 line 1 and 2 of revised version of the manuscript.

Thanks for your attention again.
Your Sincerely,
Mehmet HOROZ, MD.